Author's response to reviews

Title: A presenting with Abnormal Fingernail Beds Following Carbon Monoxide Poisoning

Authors:

Mehrangiz Hatami (mehrangizhatami@yahoo.com)
Frederick Naftolin (Frederick.Naftolin@nyumc.org)
Massoud A Khatamee (frfbaby@msn.com)

Version: 7 Date: 14 July 2013

Author's response to reviews: see over
Dear JMCR Editorial Team,

I really appreciate Editor's comments. The title of the manuscript has been changed to “A Presenting with Abnormal Fingernail Beds Following Carbon Monoxide Poisoning”. The spell of the third authors’ name also corrected, So Massoud A Khatamee in the title changed to Masood A Khatamee. The changes also have been highlighted in a yellow background in the manuscript for your consideration.